Author’s response to reviews

Title: Safety and effectiveness of vinflunine in patients with metastatic transitional cell carcinoma of the urothelial tract after failure of one cisplatin-based systemic therapy in clinical practice

Authors:

Daniel Castellano (cdanicas@hotmail.com)
Javier Puente (jpuente.hcsc@salud.madrid.org)
Guillermo De Velasco (g_develasco@yahoo.es)
Isabel Chirivella (chirivella_isa@gva.es)
Pilar López-Criado (mplopez@mdanderson.es)
Nicolás Mohedano (mohedanomohedano@yahoo.es)
Ovidio Fernández (ovidiofernandezcalvo@yahoo.es)
Icíar García-Carbonero (igcarbonero@yahoo.es)
M. Belén González (mbgonzal@hsll.es)
Enrique Grande (egrande@oncologiahrc.com)

Version: 2 Date: 9 December 2013

Author’s response to reviews: see over
Cover letter

Submission to: BMC Cancer

Title of Paper: Safety and effectiveness of vinflunine in patients with metastatic transitional cell carcinoma of the urothelial tract after failure of one cisplatin-based systemic therapy in clinical practice

Dear editor,

I hope this change will arrive promptly soon. I’m going to detail you a point-by-point description of the change made. I’ve introduced the following sentence on the Competing Interests disclosure document: “The authors declare that there are no conflicts of interest and Non-financial competing interests.”

I hope this will be enough. In other case, please it will be a pleasure to modify whatever you need.

Yours sincerely,

Daniel Castellano MD, PhD
Hospital Universitario 12 de Octubre
Avenida de Córdoba s/n
28041 Madrid
Tel No: +34913908000
Fax No: +34914603310
Email: cdanicas@hotmail.com